Author’s response to reviews

Title: Effect of children's shoes on gait: a systematic review and meta-analysis

Authors:

   Caleb Wegener (cweg6974@uni.sydney.edu.au)
   Adrienne E Hunt (adrienne.hunt@sydney.edu.au)
   Benedicte Vanwanseele (benedicte.vanwanseele@sydney.edu.au)
   Joshua Burns (joshuab2@chw.edu.au)
   Richard M Smith (richard.smith@sydney.edu.au)

Version: 4 Date: 6 January 2011

Author’s response to reviews: see over
Dear Editor,

Please find enclosed version three of our manuscript and additional files entitled “Effect of shoes on the gait of children: a systematic review and meta-analysis” following suggestions by the editor of the Journal of Foot and Ankle Research. The recommendations of the editor have been implemented.

All authors have approved the revisions to the manuscript.

Kind regards

Caleb Wegener
Adrienne Hunt
Benedicte Vanwanseele
Joshua Burns
Richard Smith